Author’s response to reviews

Title: Gender differences among Indigenous Canadians experiencing Homelessness and Mental Illness

Authors:

Brittany Bingham (bld@sfu.ca)
Akm Moniruzzaman (akm_moniruzzaman@sfu.ca)
Michelle Patterson (michelle_patterson@sfu.ca)
Jitinder Sareen (sareen@umanitoba.ca)
Jino Distasio (j.distasio@uwinnipeg.ca)
John O’Neil (joneil@sfu.ca)
Julian Somers (jsomers@sfu.ca)

Version: 4 Date: 11 Jul 2019

Author’s response to reviews:

Entered by Julian Somers (senior author)

Brittany Bingham (corresponding and first author) is on medical leave. Tables 1 and 2 have been substantially reduced and readers have been referred to a recent publication for additional details of the full sample. Minor text edits support this revision. The retention of items in Tables is related to their inclusion in the current multivariable model. I agree that the original tables were unnecessarily redundant.